Title: Hepatotoxicity induced by horse ATG and reversed by rabbit ATG: A case report

Authors:

Khalid KA Al-Anazi (khalid_alanazi@yahoo.com)
Mahmoud MD Aljurf (maljurf@kfshrc.edu.sa)
Fahad FZ Al-Sharif (alsharif@kfshrc.edu.sa)
Hamad HM Al-Omar (halomar@kfshrc.edu.sa)
Ahmed A Alami (a_alami2000@yahoo.com)
Fayyaz F Farooq (fayyazrph@hotmail.com)

Version: 2  Date: 13 April 2007

Author's response to reviews:

Assistant Editor
Journal of Medical Case Reports
Dear Sir:
Happy day to you
The required changes have been done
Many thanks